Author’s response to reviews

Title: Effect of Yerba Mate and Green Tea on Paraoxonase and Leptin Levels in Patients affected by Overweight or Obesity and Dyslipidemia: A Randomized Clinical Trial

Authors:

Guilherme Balsan (guilhermeabalsan@gmail.com)
Lúcia Pellanda (lupellanda@gmail.com)
Grasiele Sausen (grasiele.sausen@gmail.com)
Thaís Galarraga (thaisabg@terra.com.br)
Denise Zaffari (denisezaffari@gmail.com)
Bruna Pontin (brunapontin@hotmail.com)
Vera Portal (veraportal.pesquisa@gmail.com)

Version: 5 Date: 17 Dec 2018

Author’s response to reviews:

Porto Alegre, December 17rd, 2018

Dr. Sharon Kirkpatrick
Editor-in-Chief
BMC Nutrition Journal

NUTJ-D-18-00069

Effect of Yerba Mate and Green Tea on Paraoxonase and Leptin Levels in Overweight Patients With Mild Dyslipidemia: A Randomized Clinical Trial

Guilherme Balsan, Lúcia Campos Pellanda, Grasiele Sausen; Thais Galarraga; Denise Zaffari; Bruna Pontin; Vera Lúcia Portal

Dear Editors,

We appreciate the review of our paper Effect of Yerba Mate and Green Tea on Paraoxonase and Leptin Levels in Overweight Patients With Mild Dyslipidemia: A Randomized Clinical Trial, and we have made suggested modification, and it’s highlighted in yellow throughout the text.
Sincerely yours,

Dr. Vera Lucia Portal

Institute of Cardiology / University Cardiology Foundation
Avenida Princesa Isabel, 370 – 3 andar
Bairro Santana, Porto Alegre – RS, Brazil
CEP: 90.620-001
E-mail: vera.portal@hotmail.com